Author's response to reviews

Title: Protocol for a mixed methods study investigating the impact of investment in housing, regeneration and neighbourhood renewal on the health and wellbeing of residents: the GoWell programme

Authors:

Matt Egan (m.egan@sphsu.mrc.ac.uk)
Ade Kearns (a.kearns@lbss.gla.ac.uk)
Phil Mason (p.mason@socsci.gla.ac.uk)
Carol Tannahill (Carol.Tannahill@glasgow.gov.uk)
Lyndal Bond (lyndal@sphsu.mrc.ac.uk)
Jennie Coyle (Jennie.Coyle@glasgow.gov.uk)
Sheila Beck (Sheila.beck@nhs.net)
Fiona Crawford (Fiona.Crawford@glasgow.gov.uk)
Phil Hanlon (P.Hanlon@clinmed.gla.ac.uk)
Louise Lawson (l.lawson@lbss.gla.ac.uk)
Jennifer McLean (Jennifer.McLean2@glasgow.gov.uk)
Mark Petticrew (Mark.Petticrew@lshtm.ac.uk)
Elena Sautkina (elena@sphsu.mrc.ac.uk)
Hilary Thomson (hilary@sphsu.mrc.ac.uk)
David Walsh (david.walsh@drs.glasgow.gov.uk)
Glasgow Community Health and Wellbeing Programme The GoWell Programme (jennie.coyle@drs.glasgow.gov.uk)

Version: 3 Date: 19 April 2010

Author's response to reviews:

Dear Editors, thank you for your latest comments. We have made the requested changes with the following exception.

The editors' formatting comments state "Figure 1 and 2 legends: please insert these legends in a section titled 'Figure legends' after the references.” Unfortunately, we have no means of removing the legend from Figure 1 (the map of Glasgow with GoWell areas shown). It was produced for us by an external organisation and comes in a format we cannot manipulate. However, we think the legends on the image are clear and the map is useful to the article - so we hope you decide that we can retain it. We have removed the title from figure 2 but have not removed any other headings - we don't think the headings really are a 'legend' and we feel that removing or moving them would make the diagram confusing to read.

Besides the requested formatting changes, we have also made one change to the text on page 5 (under the heading GoWell). Here, we written the Programmes title out in full: "The Glasgow Community Health and Wellbeing (GoWell) Research and Learning Programme".
Best wishes, Matt Egan (on behalf of the authors)